Author's response to reviews

Title: Response rates in postal surveys of healthcare professionals between 1996 and 2005: An observational study

Authors:

Julia V Cook (J.V.F.Cook@ncl.ac.uk)
Heather O Dickinson (heather.dickinson@ncl.ac.uk)
Martin P Eccles (martin.eccles@ncl.ac.uk)

Version: 5 Date: 26 August 2009

Author's response to reviews:

Dear editors
I have completed the proof reading of the manuscript and made some minor revisions: pg 4, pg 6, pg 9, pg 10, pg 12, pg 15 and some minor revisions to the references. We are happy with this final version of the manuscript. sincerely julia cook